Author’s response to reviews

Title: Protein energy-wasting associated with nephrotic syndrome – the comparison of metabolic pattern in severe nephrosis to different stages of chronic kidney disease.

Authors:

Anna Matyjek (matyjek.anna@gmail.com)
Slawomir Literacki (sliteracki@wim.mil.pl)
Stanislaw Niemczyk (sniemczyk@wim.mil.pl)
Aleksandra Rymarz (arymarz@wim.mil.pl)

Version: 1 Date: 13 Jan 2020

Author’s response to reviews:

Dear Editor,

I have revised the manuscript according to your comments. If there are any other editorial mistakes in the manuscript or if my corrections are insufficient, I would be grateful for additional advise. I will follow your instructions immediately.

Sincerely,
Anna Matyjek